\begin{table}%t1
%\centering
\par
\caption{The {18} planet candidates with $m_{\rm 2}\sin i <11~ M_{\rm  Jup}$\ detected with {\footnotesize ELODIE} at Haute-Provence Observatory.
\label{decouvertesElodie}}       
\begin{tabular}{ll}
\hline\hline
\noalign{\vspace{0.025cm}}
{Candidate}                                      & {References} \\
\noalign{\vspace{0.025cm}}
\hline
\object{51~Peg~b}                                           & \citet{MayorQueloz95}\\
\object{14~Her~b}                                           & \citet{UdryMQ00}, \citet{NaefeloIII}\\
\object{{\footnotesize Gl}~876~b}$^{\bullet}$               & \citet{Delfosse98}\\
\object{{\footnotesize HD}~209458~b}$^{\diamond}$           & \citet{Mazeh00}\\
\object{{\footnotesize HD}~190228~b}$^{\star}$              & \citet{Sivan00} and this paper\\
\object{{\footnotesize HD}~8574~b}$^{\star}$                & this paper\\
\object{{\footnotesize HD}~50554~b}$^{\star}$$^{\dag}$      & this paper\\
\object{{\footnotesize HD}~74156~b}$^{\star}$               & \citet{NaefeloIII}\\
\object{{\footnotesize HD}~74156~c}$^{\star}$               & \citet{NaefeloIII}\\
\object{{\footnotesize HD}~80606~b}$^{\star}$$^{\circ}$     & \citet{Naef01}\\
\object{{\footnotesize HD}~106252~b}$^{\star}$$^{\dag}$     & this paper\\
\object{{\footnotesize HD}~178911~Bb}$^{\circ}$$^{\star}$   & \citet{Zucker02}\\
\object{{\footnotesize HD}~20367~b}$^{\ast}$                & \citet{Udrywash}\\
\object{{\footnotesize HD}~23596~b}$^{\ast}$                & this paper\\
\object{{\footnotesize HD}~33636~b}$^{\ast}$$^{\ddag}$      & this paper\\
\object{{\footnotesize HD}~37124~c}$^{\ast}$$^{\natural}$   & \citet{Udrywash}\\
\object{{\footnotesize HD}~150706~b}$^{\ast}$               & \citet{Udrywash}\\
\object{{\footnotesize GJ}~777~Ab}$^{\ast}$$^{\flat}$       & \citet{Udrywash}, \citet{NaefGJ777}\\
\noalign{\vspace{0.025cm}} 
\hline       
\end{tabular}       
\begin{list}{}{}
\item$^{\bullet}$\ See also \citet{Marcy98}.
\item$^{\diamond}$\ See also \citet{Charbonneau00}, \citet{Henry00}.	    
\item$^{\star}$\ Announced in {\footnotesize ESO} Press Release nb~07/01 (April~4, 2001).
\item$^{\dag}$\ Confirmed by \citet{Fischer02}.             
\item$^{\circ}$\ Variability detected by the Keck {\em G--Dwarf Planet Search}.	    
\item$^{\ddag}$\ Independently announced by \citet{Vogt02}.      
\item$^{\ast}$\ Announced during the {Scientific Frontiers in Research on 
extra-solar Planets} conference, Washington, June~2002.
\item$^{\natural}$\ See also \citet{Butler02}.
\item$^{\flat}$\ {\footnotesize HD}~190360~b.
\end{list}    
\end{table}